Author's response to reviews

Title: Socio-demographic and clinical characteristics of re-presentation to an Australian inner-city emergency department: implications for service delivery.

Authors:

Gaye GE Moore (dgmoore@optusnet.com.au)
Marie MF Gerdtz (gerdtzmf@unimelb.edu.au)
Elizabeth E Manias (emanias@unimelb.edu.au)
Graham G Hepworth (g.hepworth@ms.unimelb.edu.au)
Andrew A Dent (Andrew.Dent@svhm.org.au)

Version: 2 Date: 11 April 2007

Author's response to reviews:

Dear Editor Manager

I have made the following minor changes to the manuscript on revision prior to peer review:
1. pg 1 Ms Gaye Moore changed to Gaye Moore
2. pg 4 superscript 26-28 removed
3. pg 24 Table 2 non-homeless (n= 40,942) changed to (n = 39347)

Thank you for the opportunity to submit this manuscript. The electronic process has been easy to complete and the timely responses via email has been greatly appreciated

Yours Sincerely

Gaye Moore
The School of Nursing
The University of Melbourne
Level 1, 723 Swanston Street
Carlton, Victoria 3053 Australia